Author's response to reviews

Title: Magnetic Resonance Tracking of Transplanted Autologous Bone Marrow-Derived Mesenchymal Stromal Cell in Chronic Spinal Cord Injury at Upper Cervical Level: Case Report

Authors:

Areesak Chotivichit (areesak.cho@mahidol.ac.th)
Monchai Ruangchainikom (monchai.ortho@gmail.com)
Pipat Chiewvit (pipat8999@yahoo.com)
Adisak Wongkajornsilp (adisak.won@mahidol.ac.th)
Kittipong Sujirattanawimol (Kittipong.suj@mahidol.ac.th)

Version: 3 Date: 18 September 2014

Author's response to reviews: see over
Dear Journal of medical case reports

This is a case report of patient who received bone-marrow derived mesenchymal stem cells. There was evidence of cell migration from serial MRI; however, there was transient neurological complication from transplantation. We hope that Journal of Medical Case Reports would consider the information in this manuscript for publication.

Regards,

Monchai Ruangchainikom, MD
Department of Orthopaedic Surgery
Faculty of Medicine Siriraj Hospital, Mahidol University
monchai.ortho@gmail.com